 Vc/21-Jun-04
Week25  starting 14 June  04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 14 june ( ROCLA OK ?): 

	Bench
	Name
	Situation Mon 14June pm
	 
	-

	TBA1
	MBAR3063
	Prep 5 Pumping + Cool Down
	Q Antenna 
	 

	TBA2
	MBBL3084
	waiting for Rocla 2015 next?
	 Shafts
	 

	TBB1
	MBBL1098
	Prep 5 Pumping + Cool Down
	 
	 

	TBB2
	MBBL1120
	ICS 1 WP04 HV Test Warm
	 
	 

	TBC1
	MBAL3140
	ICS 1 WP04 HV Test Warm
	 
	 

	TBC2
	SSS 031
	PT 6.1 Training 1
	 
	 

	TBD1
	SSS11
	MM 2 LHC Cycle
	 
	 

	TBD2
	_
	_
	 
	 

	TBE1
	MBBL1119
	PT 6.4 Training 4
	 
	 

	TBE2
	MBBL3067
	ICS 4 Final connection
	 
	 

	TBF1
	MBBL1097
	ICS 2 Connect Magnet (ICS)
	Shafts?
	 

	TBF2
	MBAL2059
	ICS 5 Traveller Last
	 
	 


Test Coordinator this week: Stephane  ; MTM support on call :  (16-3602).

9 Magnets( 8 dipoles+ 1 SSS)   Disconnected after tests last week    ( Sat Midnight to Sat midnight)  
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

·  D2  preparation mechanically for SSS 
· Repair Water leak in F cluster Power Converter ( 3 days asked) Mon to Wed 
· SSS work :    AB-OP CERN Staff  daily(almost)  in new Rota 
· TO-DO List;  SSW & Surveyors Call out   
Bench status  and tests for now( Monday 14/6 at 19.00)  & next few days :  see Video Display
	Magnets under Test on 14-06 6:48:47 PM 

	-
	Bench
	Magnet
	Shafts
	Start
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	Last Test
Desc
	
Time
	
I
	
Temp.
	Total

	1
	TBA1
	MBAR3063 
	[image: image1.png]



	11-06 
	37 h.
	43 h.
	-
	-
	-
	PT 1 IAP @ cold
	14-06
(18 : 05) 
	0 A. 
	4.4 K
	80 hours

	2
	TBA2
	MBBL3084 
	[image: image2.png]



	07-06 
	22 h.
	53 h.
	44 h.
	14 h.
	29 h.
	waiting for Rocla
	14-06
(13 : 49) 
	0 A. 
	300 K
	167 hours

	3
	TBB1
	MBBL1098 
	[image: image3.png]



	11-06 
	62 h.
	15 h.
	-
	-
	-
	Prep 5 Pumping + Cool Down
	14-06
(03 : 23) 
	0 A. 
	Cooling
	77 hours

	4
	TBB2
	MBBL1120 
	[image: image4.png]



	14-06 
	51 h.
	99 h.
	-
	-
	-
	ICS 1 WP04 HV Test Warm
	14-06
(18 : 24) 
	0 A. 
	300 K
	150 hours

	5
	TBC1
	MBAL3140 
	[image: image5.png]



	14-06 
	3 h.
	-
	-
	-
	-
	ICS 2 Connect Magnet (ICS)
	14-06
(18 : 39) 
	0 A. 
	300 K
	3 hours

	6
	TBC2
	SSS 031 
	[image: image6.png]



	10-06 
	64 h.
	22 h.
	19 h.
	-
	-
	PT 6.1 Training 1
	14-06
(15 : 40) 
	12310 A. 
	1.9 K
	105 hours

	7
	TBD1
	SSS11 
	[image: image7.png]



	27-05 
	180 h.
	86 h.
	176 h.
	-
	-
	MM 2 LHC Cycle
	11-06
(19 : 07) 
	0 A. 
	1.9 K
	442 hours

	8
	TBE1
	MBBL1119 
	[image: image8.png]



	09-06 
	45 h.
	25 h.
	49 h.
	-
	-
	PT 6.5 Training 5
	14-06
(18 : 42) 
	12200 A. 
	1.9 K
	119 hours

	9
	TBE2
	MBBL3067 
	[image: image9.png]



	12-06 
	67 h.
	-
	-
	-
	-
	ICS 4 Final connection
	14-06
(16 : 16) 
	0 A. 
	300 K
	67 hours

	10
	TBF1
	MBBL1097 
	[image: image10.png]



	14-06 
	16 h.
	974 h.
	-
	-
	-
	ICS 2 Connect Magnet (ICS)
	14-06
(15 : 40) 
	0 A. 
	300 K
	990 hours

	11
	TBF2
	MBAL2059 
	[image: image11.png]



	09-06 
	45 h.
	31 h.
	27 h.
	13 h.
	2 h.
	ICS 5 Traveller Last
	14-06
(18 : 07) 
	0 A. 
	300 K
	118 hours

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR3063
	[image: image12.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL3084
	11230 Amp.
(D2-L-S01)
	12440 Amp.
(D1-L-S01)
	[image: image13.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBL1098
	[image: image14.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBL1120
	[image: image15.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAL3140
	[image: image16.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS 031
	12310 Amp.
(E_12)
	[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS11
	11758 Amp.
(No Data)
	12046 Amp.
(I_12)
	12067 Amp.
(I-34)
	12423 Amp.
(E-12)
	12140 Amp.
(E-34-S02)
	12568 Amp.
(E 12-S06)
	12778 Amp.
(E-34-S06)
	12856 Amp.
([image: image18.jpg]


)
	-
	-
	-
	-
	-

	8
	MBBL1119
	11350 Amp.
(D2-U)
	11476 Amp.
(D1_L)
	11821 Amp.
(D1-L)
	12002 Amp.
(D2-U)
	12200 Amp.
(D1-L)
	[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-

	9
	MBBL3067
	[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBL1097
	[image: image21.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL2059
	11638 Amp.
(D2_L)
	11836 Amp.
(D2_L)
	12006 Amp.
(D2-L)
	12404 Amp.
(D2-U)
	12749 Amp.
(D2-l)
	12850 Amp.
([image: image22.jpg]


)
	-
	-
	-
	-
	-
	-
	-


Results by Week

	Magnets under Test from 06-06-2004 to 12-06-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBBL3122 
	[image: image23.png]



	01-06-2004 
	51 h.
	45 h.
	32 h.
	17 h.
	16 h.
	11-06-2004 
	240 hours
	
	

	2
	TBA2
	MBAL3161 
	[image: image24.png]



	31-05-2004 
	59 h.
	30 h.
	32 h.
	18 h.
	28 h.
	07-06-2004 
	168 hours
	
	

	3
	TBB1
	MBAR2056 
	[image: image25.png]



	02-06-2004 
	41 h.
	34 h.
	25 h.
	15 h.
	30 h.
	11-06-2004 
	216 hours
	
	

	4
	TBB2
	MBBL1106 
	[image: image26.png]



	02-06-2004 
	18 h.
	23 h.
	21 h.
	17 h.
	42 h.
	08-06-2004 
	144 hours
	
	

	5
	TBC1
	MBBL2082 
	[image: image27.png]



	02-06-2004 
	33 h.
	27 h.
	28 h.
	19 h.
	5 h.
	08-06-2004 
	144 hours
	
	

	6
	TBC2
	SSS21 
	[image: image28.png]



	13-05-2004 
	14 h.
	82 h.
	391 h.
	7 h.
	112 h.
	07-06-2004 
	600 hours
	
	

	7
	TBE1
	MBAL3157 
	[image: image29.png]



	29-05-2004 
	74 h.
	44 h.
	51 h.
	27 h.
	46 h.
	09-06-2004 
	264 hours
	
	

	8
	TBE2
	MBAL1086 
	[image: image30.png]



	03-06-2004 
	45 h.
	41 h.
	50 h.
	15 h.
	16 h.
	11-06-2004 
	192 hours
	
	

	9
	TBF2
	MBBL1117 
	[image: image31.png]



	03-06-2004 
	46 h.
	32 h.
	22 h.
	15 h.
	17 h.
	09-06-2004 
	144 hours
	
	


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL3122
	12154 A.
(D2_U)
	12398 A.
(D2_U)
	12555 A.
(D1-L)
	12692 A.
(D1-U)
	12757 A.
(D1-U)
	12798 A.
(D1-L-S01)
	[image: image32.png]



	-
	-
	-
	-
	-
	-

	2
	MBAL3161
	10694 A.
(D1-U)
	12045 A.
(D1-L)
	[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBAR2056
	11723 A.
(D2_L)
	12650 A.
(D2-U)
	[image: image34.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBL1106
	12101 A.
(D1 L)
	12537 A.
(D2-L)
	[image: image35.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBL2082
	11641 A.
(No Data)
	12265 A.
(D1 U)
	[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS21
	12003 A.
(QE-34)
	12105 A.
(QE-12)
	12458 A.
(No Data)
	[image: image37.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBAL3157
	9662 A.
(D1-U)
	10791 A.
(No Data)
	12084 A.
(D2_L)
	12570 A.
(D2_U)
	12678 A.
(D2 - U)
	12795 A.
(D1_L)
	[image: image38.png]



	-
	-
	-
	-
	-
	-

	8
	MBAL1086
	11122 A.
(D2_U)
	11468 A.
(D2-U)
	11818 A.
(D1-U)
	12031 A.
(D1_L)
	12338 A.
(D1-L)
	12672 A.
(D2-U)
	12772 A.
(D1-L)
	12859 A.
(D2-U)
	12850 A.
[image: image39.jpg]



	-
	-
	-
	-

	9
	MBBL1117
	11105 A.
(D2_U)
	12050 A.
(D2-L)
	[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


· General Problems last week :  
	Date
	Bench
	Problem
	Type
	Application/soft
	Action/advise

	Thursday _10_A
	TBA2
	Trigger on 2 potaim cards V_clamp_ext and V_supra_ext because Vtap  lost during the cool down.


	MTM
	
	Ewald called. Following the advise of Giorgio a bridge on 7-8 cards on the double output was placed.

	Thursday _10_N
	TBA2
	 The strap was not put in teh wrong pins. It triggers the potaim cards 5-6 .


	MTM
	
	A.Siemko corrected the position of the strap of the double output. Then OK.



	Thursday _10_N
	TBA2
	MM position : Wrong setting (3117 instead of 3084)
	MTM
	MMP
	Ewald corrected the setting.

	Thursday _10_N
	TBB2
	Tcoil indicates low temperatures for few sectors. Values were correct in the Anti-cryostat display.


	MTM
	
	

	Friday _11_A
	TBD1
	SSS11: SPA during LHC cycle.


	MTM
	
	Guy called and solved the problem. Quench and LHC cycle repeated.

	Friday _11_A
	TBC1
	During IAP at cold, compensation of AC signals difficult because of noise in some Voltages. 


	MTM
	IAP
	Guy and Richard managed to reduce the noise for signals less than 5 V, not for the other ones.

	Saturday 12 _A
	TBA2
	MM on 3081: Problem in rotation on shaft 24. RMS rotation speed >3%.


	MTM
	
	Stephane called. He checked the data and observed the rotation during the beginning of load-line.

RMS degraded so he decided to stop the measurements.

During the removal of the shaft the glue between the two ceramic sectors broke.

	Saturday 12 _A
	TBA1
	T of anticryostat not correct. The gamelle number not found. 


	MTM
	
	GHH took the number in the box  used for the connexion.

	Saturday 12 _N
	TBA1
	IAP at warm: T sensor gave wrong temperature 87 K instead of 300 K. 


	MTM
	
	Stephane and Guy informed Sunday night. Stephane asked to check R sensor and the coefficient in MTF.

	Saturday 12 _N
	TBA1
	IAP at warm: Mag_300_X1 and Mag 300 air_X1 not OK. 


	MTM/ACR
	IAP
	Guy informed. Cryo problem.

	Sunday 13 _M
	TBE1
	Valve of water not working in automatic way.  


	MTM
	
	Switch in manual mode. Valve moved by hand.

	Sunday 13 _A
	TBE1
	HF quench at 1.5 kA: Timing error.

LF quench at 1.5 kA: QH not discharged.


	MTM
	hf, lf
	Stephane called: He checked the signals. Give the OK to continue. The propagation of quench is too slow and pre-trigger time is not enough (3s).

	Sunday 13 _N

Monday_14_M
	TBC2
	SSS HV test:  Orbit dipole vs ground failed because the leads are freezing. 

IAP: AC compensation difficult to perform because AC source got loaded (4 volt cannot be maintained).
	MTM

MTM
	
	Giorgio checked with the OP team the compensations. OK

ON the scope the option -20 DB/decade was activated (too low gain)?








